Author's response to reviews

Title: Virtual Field Trips as physically active lessons for children: A pilot study

Authors:

Emma Norris (e.norris.11@ucl.ac.uk)
Nicola Shelton (n.shelton@ucl.ac.uk)
Sandra Dunsmuir (s.dunsmuir@ucl.ac.uk)
Oliver Duke-Williams (o.duke-williams@ucl.ac.uk)
Emmanuel Stamatakis (emmanuel.stamatakis@sydney.edu.au)

Version: 2 Date: 29 October 2014

Author's response to reviews:

Title page, consent details and line numbers added